**Author's response to reviews**

**Title:** Gene Expression Meta-Analysis Supports Existence of Molecular Apocrine Breast Cancer with a Role for Androgen Receptor and Implications Interactions with ErbB Family

**Authors:**

- Sandeep Sanga (sandeep.sanga@mail.utexas.edu)
- Bradley M Broom (bmbroom@mdanderson.org)
- Vittorio Cristini (vittorio.cristini@uth.tmc.edu)
- Mary E Edgerton (medgerton@mdanderson.org)

**Version:** 5  **Date:** 10 August 2009

**Author's response to reviews:** see over
August 10, 2009

Dear Dr. Burgess,

Thank you for your letter indicating that the manuscript has been accepted. As requested, we added a brief description of our methods to the Abstract and we re-arranged the it into four sections titled Background, Methods, Results and Conclusions as compared with our previously submitted version that has three sections titled Background, Results, and Conclusions. Minor adjustments were made to the other sections to remove redundant information and to keep the abstract limited to less than 350 words.

Our only other change has been to format the paper in keeping with the format of your published papers. We note that we renamed our supplemental files as additional files after the format of your journal. We also insured that they are referred to properly in the text. There are otherwise no changes to the text of the manuscript.

Thank you.

Sincerely,

Mary E. Edgerton

For

Sandeep Sanga
Bradley M Broom
Vittorio Cristini
Mary E. Edgerton